Author's response to reviews

Title: Attenuating effect of Standardized Fruit Extract of Punica Granatum in rat model of Tibial and Sural Nerve Transection induced neuropathic pain

Authors:

Vivek Jain Mr (vivek19j@gmail.com)
Ashutosh Pareek Mr (ashu83aadi@gmail.com)
Yashumati Ratan Ms (yashu.bu@gmail.com)
Nirmal Singh Dr (nirmal_puru@rediffmail.com)

Version: 2 Date: 6 April 2013

Author's response to reviews:

Response to Comments

1. Please include Author's Contributions in the manuscript.
Response: Authors contributions is included in the text [highlighted in yellow].

2. Please include a Competing Interests section in your manuscript.
Response: competing interests section has now been incorporated in the text [highlighted in yellow].